Author's response to reviews

Title: Knowledge of modifiable risk factors of heart disease among patients with acute myocardial infarction in Karachi, Pakistan: a cross sectional study

Authors:

Muhammad S Khan (muhammad.saleem@aku.edu)
Fahim H Jafary (fahim.jafary@aku.edu)
Tazeen H Jafar (tazeen.jafar@aku.edu)
Azhar M Faruqui (nicvdedo@khi.comsats.net.pk)
Syed I Rasool (ssrasool@hotmail.com)
Juanita Hatcher (juanita.hatcher@aku.edu)
Nish Chaturvedi (n.witt@imperial.ac.uk)

Version: 5 Date: 15 April 2006

Author's response to reviews: see over
Dear Sir/Madam,

Thanks for your valuable revision of the paper.

I have addressed the comments both by providing with a point-by-point response to them and by revising the manuscript.

**Reviewer: Julie J Zerwic**

Major Compulsory Revisions (that the author must respond to before a decision on publication can be reached)

-----------------------------------------------------------------------------------

Minor Essential Revisions (such as missing labels on figures, or the wrong use of a term, which the author can be trusted to correct)
Page paragraph line

2 1 3 Delete the word "good" before level.

2 1 3 Thank you. Word good has been deleted before level

2 2 5 Delete the word "about" before three.

2 2 6 word “about” has been deleted before three

2 2 5 Change participant to participants.

2 2 6 participant has been changed to participants

5 1 5 put "the" between often and source.

5 1 7 “the” has been added between often and source.

5 1 6 put "as" before updated.

5 1 9 “as” has been added before updated.

5 2 7 Replace the sentences from "there is..." with the following. "The purpose of this study was to examine knowledge of heart disease risk in a sample of individuals who were hospitalized with their first acute myocardial infarction (AMI). This was part of a larger study focused on delay in seeking early medical care after AMI onset."

5 2 6 sentence has been replaced with following sentence

The purpose of this study was to examine the level of knowledge of modifiable risk factors of heart diseases in a sample of individuals who were hospitalized with their first
acute myocardial infarction (AMI). This was part of a larger study focused on delay in seeking early medical care after AMI onset

7 2 4 Delete sentence since it repeats previous sentence.
7 2 4 sentence has been deleted

8 3 5 Delete sentence since it repeats previous sentence.
8 2 5 sentence has been deleted

12 2 9 Delete sentence since it repeats previous sentence.
12 2 9 sentence has been deleted

15 2 1 Change awareness to aware.
15 2 1 awareness has been changed to aware

Thank you for your consideration.

Sincerely,

Tazeen H. Jafar, MD MPH
Director, Clinical Epidemiology Unit
Assistant Professor
Medicine and Community Health Sciences
The Aga Khan University, Karachi, Pakistan
Phone: 92-21-4930051 ext 4812
FAX: 92-21-493-4294
Email: tazeen.jafar@aku.edu